Author’s response to reviews

Title: High prevalence of diarrheagenic Escherichia coli carrying toxin-encoding genes isolated from children and adults in southeastern Brazil

Authors:
Liliana Spano (liliana.spano@ufes.br)
Keyla Cunha (fonseca.keyla@gmail.com)
Mariane Monfardini (marianemonfardini@yahoo.com.br)
Rita de Cássia Fonseca (fonseca.vix@terra.com.br)
Isabel Cristina Scaletsky (scaletsky@unifesp.br)

Version: 5 Date: 29 Nov 2017

Author’s response to reviews:

November, 2017

Journal Editorial Office
BMC Infectious Diseases

Dear Edward Spofford,

Enclosed you will find the revised manuscript entitled “High prevalence of diarrheagenic Escherichia coli carrying toxin-encoding genes isolated from children and adults in southeastern Brazil”.

Editor Comments:

1. Ethics statement was clarified in the revised manuscript (page 13, lines 306-309).

2. Declarations section as currently the Funding, Ethics approval, Consent to participate, and Consent for publication were included in the revised manuscript.

Thank you very much for your consideration

Sincerely,
Isabel Scaletsky